

Complexity of Infection and Genetic Diversity in Cambodian Plasmodium vivax

Lindsey Friedrich, Jean Popovici, Saorin Kim, Lek Dysoley, Peter A Zimmerman, Didier Ménard, David Serre

▶ To cite this version:

Lindsey Friedrich, Jean Popovici, Saorin Kim, Lek Dysoley, Peter A Zimmerman, et al.. Complexity of Infection and Genetic Diversity in Cambodian Plasmodium vivax. PLoS Neglected Tropical Diseases, 2016, 10 (3), pp.e0004526. 10.1371/journal.pntd.0004526 . hal-02560632

HAL Id: hal-02560632 https://hal.science/hal-02560632v1

Submitted on 11 May 2020

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

CrossMark

Citation: Friedrich LR, Popovici J, Kim S, Dysoley L, Zimmerman PA, Menard D, et al. (2016) Complexity of Infection and Genetic Diversity in Cambodian *Plasmodium vivax*. PLoS Negl Trop Dis 10(3): e0004526. doi:10.1371/journal.pntd.0004526

Editor: Jane M. Carlton, New York University, UNITED STATES

Received: November 12, 2015

Accepted: February 18, 2016

Published: March 28, 2016

Copyright: © 2016 Friedrich et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The sequence data are available in NCBI SRA under the BioProject PRJNA295043.

Funding: This work was funded by a National Institutes of Health NIAID award to DS (R01 A103328). LRF was partially supported by a Case Western Reserve University training grant in Geographic Medicine (T32 A107024). Additional support was provided by a Global Fund Grant Malaria Program Round 9 award to DM (CAM-S10-G14-M). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Complexity of Infection and Genetic Diversity in Cambodian *Plasmodium vivax*

Lindsey R. Friedrich¹, Jean Popovici², Saorin Kim², Lek Dysoley³, Peter A. Zimmerman⁴, Didier Menard², David Serre¹*

 Genomic Medicine Institute, Cleveland Clinic Lerner Research Institute, Cleveland, Ohio, United States of America, 2 Unite d'Epidemiologie Moleculaire, Institut Pasteur du Cambodge, Phnom Penh, Cambodia,
National Center for Parasitology, Entomology and Malaria Control, Phnom Penh, Cambodia, 4 Center for Global Health and Diseases, Case Western Reserve University, Cleveland, Ohio, United States of America

* serred@ccf.org

Abstract

Background

Plasmodium vivax is the most widely distributed human malaria parasite with 2.9 billion people living in endemic areas. Despite intensive malaria control efforts, the proportion of cases attributed to *P. vivax* is increasing in many countries. Genetic analyses of the parasite population and its dynamics could provide an assessment of the efficacy of control efforts, but, unfortunately, these studies are limited in *P. vivax* by the lack of informative markers and high-throughput genotyping methods.

Methodology/Principal Findings

We developed a sequencing-based assay to simultaneously genotype more than 100 SNPs and applied this approach to ~500 *P. vivax*-infected individuals recruited across nine locations in Cambodia between 2004 and 2013. Our analyses showed that the vast majority of infections are polyclonal (92%) and that *P. vivax* displays high genetic diversity in Cambodia without apparent geographic stratification. Interestingly, our analyses also revealed that the proportion of monoclonal infections significantly increased between 2004 and 2013, possibly suggesting that malaria control strategies in Cambodia may be successfully affecting the parasite population.

Conclusions/Significance

Our findings demonstrate that this high-throughput genotyping assay is efficient in characterizing *P. vivax* diversity and can provide valuable insights to assess the efficacy of malaria elimination programs or to monitor the spread of specific parasites.

Author Summary

Plasmodium vivax is responsible for most malaria cases outside Africa but remains poorly understood. Here we describe a high-throughput assay that enables genotyping more than

Competing Interests: The authors have declared that no competing interest exist.

100 SNPs in 96 samples simultaneously. We applied this assay to characterize *P. vivax* genetic diversity in ~500 individual infections collected throughout Cambodia and including multiple time points. Our analyses revealed that the Cambodian *P. vivax* population is very diverse genetically but geographically homogenous, without any obvious boundaries. We also observed that almost all infected individuals carried more than one *P. vivax* parasite, though the proportion of single parasite infections tends to increase in recent years. Our study illustrates how genetic information can be generated to monitor temporal and geographical variations in parasite populations and help assess the efficacy of malaria control programs.

Introduction

Plasmodium vivax is the most widely distributed human malaria parasite, threatening almost 40% of the world population [1]. It is the major cause of malaria outside of Africa, with up to 390 million clinical infections each year, and is responsible for long-term chronic illness that has dramatic consequences for the health and economy of endemic regions. Despite the implementation of extensive control efforts, the proportion of cases attributed to P. vivax has significantly increased over time compared to *P. falciparum* [2]. For example, in Cambodia the malaria control activities in the last ten years-extended access to insecticide treated bed nets, distribution of rapid diagnostic tests (RDTs) to health facilities, switch to dihydroartemisininpiperaquine as first line therapy for treatment of uncomplicated malaria-have led to a 50% decrease in malaria cases (from 113,855 cases in 2004 to 56,271 cases according to data from the Ministry of Health). However, these measures disproportionally affected falciparum malaria: for the first time in 2011, the number of P. vivax infections (42,901) became higher than the number of *P. falciparum* infections (33,326) and by 2014, *P. vivax* was detectable in 78% of the malaria cases (46% of mono-infections and 32% of mixed infections). In order to effectively design elimination strategies to combat P. vivax, it is essential that we better understand how the parasite population is changing in response to control measures. This is particularly important for P. vivax since our inability to culture the parasite in vitro [3] complicates measuring the efficacy of antimalarial drugs in the population (though ex vivo studies can partially overcome this limitation [4-6]). Unfortunately, the lack of an *in vitro* culture system also limits the amount of genetic and genomic data available for P. vivax. Most population genetic studies in *P. vivax* have relied on small numbers of microsatellite markers (e.g., see [7-9]) or single nucleotide polymorphisms (SNPs) [10,11] or targeted a few protein coding genes [12]. These studies therefore have limited power to identify population structure (that typically requires many unlinked markers) or to detect the presence of multiple, genetically-different parasites in a single infection (later referred to as complexity of infection or COI). Whole genome sequencing studies have the potential to circumvent these limitations but i) are prohibitively expensive for analyzing hundreds of samples and ii) are hindered by the presence of the patient DNA and require processing the samples before sequencing (e.g., using CF11 columns $[\underline{13}]$ or sequence capture $[\underline{14}]$).

We describe here a novel genotyping assay that provides sequence information from over 100 selected polymorphic loci and allows robust assessments of *P. vivax* population genetic diversity and COI. This technique enables high-throughput genotyping of parasite DNA without being hampered by human contamination (as *P. vivax* DNA is specifically amplified), provides a quantitative assessment of the proportion of each allele at the targeted SNPs, and can be easily customized to any parasite population or research question (as different markers can be

investigated by simply changing the PCR primers). We demonstrate the applicability of this approach by analyzing close to 500 *P. vivax* infections collected throughout Cambodia. Our analyses show that the data generated enable the characterization of genetic diversity and population stratification in Cambodian *P. vivax*, and the assessment of COI in this region, as well as its main determinants.

Materials and Methods

Ethics Statement

This study was conducted in accordance to the Institutional Review Board (IRB) National Ethics Committee for Health Research of Cambodia (IRB 038NECHR) and the Cleveland Clinic IRB (IRB 12–936). All patients provided written informed consent for the collection of blood samples, with the understanding that samples would be de-identified before processing.

Samples

We collected blood samples from 401 *P. vivax* symptomatic patients from nine locations across Cambodia (Fig 1) at three time points (in 2004, 2011 and 2013) and 88 asymptomatic patient



Fig 1. Map showing the approximate location of the recruitment sites in Cambodia. The diagonal line represents a no/low transmission zone and is used in our analyses to separate sites between western and eastern Cambodia. Numbers in parentheses indicate the number of samples included in the final analyses for each location.

doi:10.1371/journal.pntd.0004526.g001

samples from one district at two time points (2012 and 2013). Note that different individuals were recruited at each time (i.e., no repeated sampling of the same individuals over time). Demographic and clinical data on the samples analyzed are provided in <u>S1 Table</u>.

For each patient, we collected blood spots from finger prick or 5 mL of fresh blood in EDTA vacutainers and extracted DNA using a Qiagen Blood Mini Kit following the manufacturer's protocol.

Amplification by Locus-Specific PCR and Library Preparation

We selected 128 SNPs among those for which both alleles were observed in at least two Cambodian infections that had been whole genome sequenced [15]. This condition was chosen to increase the chance that these SNPs were variable among Cambodian *P. vivax*. We used Primer3 v4.0 [16] to design primers targeting 100–300 bp around each selected SNP (S2 Table). Each primer included a modified 5' tail to allow barcoding and incorporation of the Illumina sequencing primers by a second PCR (Fig.2). We pooled 10 mM of primers from eight to ten primer pairs (according to their melting temperature) and generated a total of 13 primer pools.

For symptomatic patient samples, we randomly assigned DNA into five 96-well plates and included duplicates for four samples as well as 25 water controls. For each plate, we conducted 13 individual PCR reactions (i.e., one primer pool per PCR reaction) to amplify all 128 SNPs using for each PCR, 1 μ l of DNA (30–50 ng DNA template) in a 50 μ l reaction consisting of 10 mM of each dNTP, 4 μ l of 25 mM MgCl₂, 10 μ l 5X GoTaq Flexi buffer, 10 μ l of primer pool (containing 10 μ M of each primer), and 1 unit of GoTaq DNA Polymerase. Reactions consisted of an initial denaturation of 3 min at 94°C followed by 30 cycles of 45 sec at 94°C, 45 sec at



Fig 2. Overview of the genotyping assay. The grey bar represents the targeted DNA sequence containing a known SNP. A first PCR (*a*) amplifies 100–300 bp of target DNA sequence using locus-specific primers (in red) including a 5'end tail (in blue). Note that this first PCR is typically performed in multiplex with 8–10 primer pairs in the same amplification reaction. After pooling the amplified products from 13 independent multiplex PCRs, the molecules are amplified by a second PCR (*b*) targeting the 5'end tail. This PCR adds to each amplified molecule a oligonucleotide index specific to each sample (grey box) and the Illumina sequencing primers. The pools are then sequenced (*c*) to generate an index sequence and two overlapping reads.

doi:10.1371/journal.pntd.0004526.g002

56°C, and 45 sec at 72°C, and a final 3 min extension at 72°C. PCR amplification products were visualized on 1% agarose mini-gels stained with ethidium bromide.

We followed an identical procedure and randomly assigned asymptomatic patient samples into one 96-well plate containing three positive (DNA from symptomatic samples) and five negative (water) controls. For each plate we conducted 13 PCR reactions as described above to amplify all 128 SNPs, but using 45 cycles and Phusion High Fidelity Taq reagents.

We combined the 13 PCR products from each individual sample and purified them using a QIAquick 96 PCR Purification Kit. We then performed a second PCR to add to each PCR product i) Illumina adapters and ii) a 6-nucleotide sequence specific to each individual sample (Fig 2). PCR conditions were identical to those reported above except for the number of cycles (only 10 cycles were performed in this second PCR) and the primer concentration (1 μ l of each primer at 10 μ M). We then pooled together the barcoded PCR products from 96 samples into a single tube and purified them using a Zymo DNA Clean and Concentrator kit. To remove primer dimers, we excised and purified bands between 200 and 600 bp from a 2% agarose gel using a Zymoclean Gel Extraction kit. Finally, we verified the library quality and quantity by Agilent Bioanalyzer-2100 and sequenced each library (containing all PCR products from 96 samples) on an Illumina MiSeq to generate 11–16 million 250 bp paired-end reads.

Sequence Analysis

We parsed all reads generated according to their respective 6-mer sequences and collapsed the paired-ends of each read into a single consensus sequence using PANDAseq [17]. We then mapped all consensus sequences to the *P. vivax* Salvador I strain reference genome sequence [18] using Bowtie2 v2.2.5 [19].

We eliminated from our analyses any targeted locus for which some of the amplified sequences mapped to more than one genome location, as this could indicate amplification of paralogous sequences. We also removed from further analyses loci that displayed more than two alleles at any given nucleotide position as this may also indicate amplification of paralogous sequences and, even if these alleles are genuine, would violate a common assumption of population genetic analyses (the infinite site model). Finally, we discarded loci for which different alleles show systematic differences in coverage (e.g., if across all infections, haplotype A was always represented by more reads than haplotype B) as this could indicate allelic dropout and poor amplification (e.g., the existence of polymorphisms at the primer sites). For symptomatic infections, we restricted our analyses to loci covered at ≥ 100 X in at least 100 samples. Individuals having less than 50 loci successfully genotyped based on the previous criteria were excluded. For asymptomatic infections, we included all individuals successfully genotyped (≥ 100 X) at least 1 SNP.

For each sample and each targeted SNP, we determined the coverage, and the frequencies of the reference and alternative alleles using samtools mpileup. Additionally, we determined the haplotypes at each locus for each sample by considering the combination of alleles present on the same read using the 'MD' string of the alignment files that summarizes all mismatch positions between the aligned reads and the reference genome. For both the targeted SNP genotypes and the locus haplotypes, we only considered alleles present in at least 10% of the reads of a given sample to avoid including PCR and sequencing errors.

Complexity of Infection

We classified samples based on their detected clonality: samples that carried a single allele at all positions (i.e., the minor allele at any given nucleotide was never observed in >10% of the reads) were classified as monoclonal, while all other samples were considered polyclonal. For

some analyses, we included, together with monoclonal infections, samples that displayed a single polymorphic site (i.e., only one nucleotide position showed two alleles) as for these infections we can reliably determine the combination of unlinked alleles carried by one parasite at all positions (i.e., the "phased" genotypes). We referred to these infections as biclonal.

We determined the combination of all major alleles across all initially targeted SNPs for the symptomatic infections (i.e., their "barcodes" [10]), to assess how informative these markers were at differentiating infections and calculated the mean number of nucleotide substitutions among them using MEGA v.6.0 [20]. For these analyses, we only considered monoclonal and biclonal infections.

We also determined the most likely number of clones present in each symptomatic infection using genotypes at the initially targeted SNPs and a maximum likelihood approach similar to the one implemented in COIL [21]: we estimated the population allele frequency at each initially targeted SNP using the mono- and biclonal infections and calculated, for each infection, the likelihood of the observed genotypes (i.e., either only the reference allele, only the alternative allele or both alleles at each SNP) when the infection contained 1 to 5 unrelated parasites. We thus estimated the most likely number of clones present in each sample. Note that the reference allele frequencies are very similar between the mono- and biclonal infections on one hand and the polyclonal infections on the other hand (S1 Fig) confirming that all these parasites belong to the same population.

We evaluated the correlations between different demographic and clinical parameters and COI using i) the assessment of monoclonality based on the presence of more than one allele at any sequenced locus and ii) the most likely number of clones in each infection. We tested the influence of geography (western vs. eastern Cambodia, using the low transmission zone as the boundary), age (<16 years of age vs. \geq 16 years), sex (male vs. female), sampling years (2004, 2011, and 2013), glucose-6-phosphate dehydrogenase (G6PD) deficiency (according to reported World Health Organization guidelines), hemoglobin (Hb) genotype, and hemoglobin concentration (in g/dl) [22]. We also evaluated the effect of parasitemia (measured in parasites/µl) on COI and whether parasitemia itself was influenced by the other demographic and clinical parameters.

Finally, we tested whether the COI differed between symptomatic and asymptomatic infections. To account for the low genotyping success in asymptomatic samples, we randomly selected, for each asymptomatic sample, one symptomatic infection (from the same location and year) and used only the genotypes successfully typed in the asymptomatic sample to assess its clonality. We then compared the number of monoclonal infections in asymptomatic samples to the number observed in the randomly matched symptomatic samples. We repeated this procedure 1,000 times and calculated how often we observed more polyclonal infections in the symptomatic subsamples than in the asymptomatic infections.

Population Stratification

We assessed population stratification in Cambodian *P. vivax* with the program STRUCTURE v2.3.4 [23] and the entire haplotype information at the targeted loci using all symptomatic infections determined to be either mono- or biclonal. To increase the discriminating power of our analyses and facilitate the interpretation of the results, we included genotypes at the same loci inferred from whole genome sequencing data generated from parasites collected in Thailand (N = 9) [24] and from non-Asian locations (N = 11) [15,25–27]. We assigned the parasites into K = 1 to K = 4 populations using 100,000 burn-ins, 1,000,000 Markov chain Monte Carlo (MCMC) steps and a model of correlated allele frequency with admixture. For each *K*, we compared the results from five independent runs to verify convergence to a similar solution and recorded the proportions of ancestry of each parasite.

We then tested whether the coefficients of ancestry of each parasite were significantly associated with geography (western vs. eastern Cambodia and between countries), age (child (<16 years of age) vs. adult (\geq 16 yrs.)), and sex (male vs. female) using *t*-tests. An ANOVA was used to evaluate differences among sampling years.

Accession Numbers

The sequence data are available in NCBI SRA under the BioProject PRJNA295043

Results

High-Throughput Genotyping

We developed a multiplex genotyping assay (Fig 2) that targeted 128 loci of 100–300 bp distributed throughout the *P. vivax* genome and covering all chromosomes (Fig 3). The loci were selected from a collection of SNPs found to be polymorphic among Cambodian *P. vivax* isolates that have been sequenced for their entire genome [15]. We applied this approach to analyze 489 *P. vivax*-infected individuals from Cambodia (401 symptomatic and 88 asymptomatic individuals). Out of the 128 primer pairs tested, one primer pair amplified DNA sequences that mapped to two different regions of the genome and was removed from the analyses. We also discarded one primer pair that amplified a low complexity locus causing misalignments in a homopolymer region. We additionally discarded six loci that displayed three alleles at some nucleotide positions (possibly indicating the amplification of unannotated paralogous loci) and 20 other loci that were not sequenced at 100 X in at least 100 symptomatic patient samples. The complete list of loci, their amplification primers and their inclusion in the final analysis is displayed in <u>S2 Table</u>. Overall, out of the 128 targeted loci, 100 were used in the final analysis.



doi:10.1371/journal.pntd.0004526.g003

Four samples were analyzed in duplicate and showed very similar genotyping results (<u>S2 Fig</u>) suggesting that this approach did not generate many false positive calls.

Of the 401 initial symptomatic patient samples, 99 samples were excluded from our analyses since we could only reliably genotype them (i.e., sequenced by >100 reads at each locus) at less than 50 SNPs. Note that the genotyping success of symptomatic samples was not associated with parasitemia (p = 0.23), geographic origin (p = 0.21), or time of collection (p = 0.38). The samples included in the final analyses comprised infections from nine districts across Cambodia and three time points (Fig 1, S1 Table). The genotyping of asymptomatic patients was less successful and, out of the 88 initial samples, 43 were excluded from further analysis since they did not have 100 X read coverage at ≥ 1 SNP.

Rigorous Identification of Complex Infections

P. vivax is haploid during the blood stage infection and monoclonal infections should therefore display a single allele at each nucleotide position. Using genotypes of the 100 SNPs that passed quality filters, 152 symptomatic patient samples, out of the 302 infections analyzed (50.3%), showed two alleles at, at least, one SNP, indicating the presence of two or more clones (i.e., polyclonal infections). For the remaining 150 samples (49.7%), the 100 SNPs all displayed a single allele and we were not able to rule out that these infections were monoclonal. Note that to limit the effect of amplification or sequencing errors, only alleles present in at least 10% of the reads were analyzed and therefore our analyses did not consider very minor clones.

Since our genotyping assay relied on high-throughput sequencing of 100–300 bp surrounding each targeted SNP, it enabled discovery and genotyping of additional DNA polymorphisms located in proximity to the targeted SNP position. In the 100 loci successfully analyzed in the 302 symptomatic samples, we identified a total of 274 SNPs. Using all 274 SNPs, only 25 infections (8%) were monoallelic at each nucleotide position (i.e., "monoclonal") while 277 (92%) samples showed clear evidence of polyclonality. Of the latter, 60 infections (22%) displayed two alleles at a single locus. We referred to these samples as "biclonal" and included them with monoclonal infections in a subset of our analyses since they enable inferring a "phased" multilocus haplotype for these samples (i.e., determining which alleles are carried by the same clone).

To assess how many genetic markers were necessary to rigorously identify COI, we randomly sub-sampled targeted SNPs and entire loci, and determined the proportion of infections deemed to be polyclonal using these data. Our results showed that single markers have limited power to identify COI and that more loci sequenced revealed more polyclonal infections, without apparent saturation (Fig 4).

We also estimated the most likely number of clones present in each infection using a maximum likelihood approach similar to the one implemented in COIL [21]. On average, each Cambodian infection contained 1.81 clones with 1.65% of the infections displaying more than three clones. Note that this analysis relied only on the 100 initially targeted SNPs and therefore underestimated the true COI (S3 Fig).

Determinants of the Complexity of Infection in Cambodian P. vivax

We then investigated which parameters influenced COI in Cambodian *P. vivax* infections by i) comparing monoclonal and polyclonal infections (as determined by the genotypes at the 274 SNPs identified in the 100 loci sequenced), and ii) comparing infections with different numbers of clones (inferred statistically from the genotypes at the 100 unlinked SNPs). Using both data sets, we found that COI did not statistically differ between infections from western and eastern Cambodia (Table 1). Similarly, we found that COI was not significantly associated with



Fig 4. Proportion of monoclonal infected detected as a function of the number of markers genotyped. The black circles indicate the percentage of infections determined to be monoclonal using only single SNPs (i.e., the initially targeted SNPs) while the grey diamonds show the results using the entire locus sequences.

doi:10.1371/journal.pntd.0004526.g004

parasitemia or the age or sex of the patient (Table 1). We also tested the association between various red blood cell phenotypes and COI but found no influence of hemoglobin level, hemoglobin genotype or G6PD deficiency (Table 1). Separating G6PD deficiency in mild and severe deficiencies, or restricting the analysis to male patients only (that are haploid for G6PD) did not change the results (p>0.14 and p>0.49, respectively). Note that, in this cohort, the parasitemia was not influenced by any of these parameters (Table 1).

By contrast, we identified a significant association between COI and the time of collection (p = 0.03 and p = 0.05, respectively, for the comparison between monoclonal and polyclonal infections and the correlation between the year of collection and the most likely number of clones). The symptomatic infections collected in more recent years displayed significantly lower COI than older infections (Fig 5). The same pattern was observed by analyzing this association separately for each location (S4 Fig).

We then tested whether COI statistically differed between symptomatic and asymptomatic infections collected at the same location and the same time. To account for differences in

Table 1. Association between clinical and demographic parameters and complexity of infection. The table shows the significance of the association (*p*-values) when comparing monoclonal to polyclonal infections (% monoclonal) or testing the correlation with the most likely numbers of clones of the infections (# clones). The association of the same parameters with parasitemia is also indicated.

| | Parasitemia | % monoclonal | # clones |
|---------------------------|-------------|--------------|----------|
| Sex | 0.54 | 0.95 | 0.37 |
| Age (< >16 years old) | 0.85 | 0.98 | 0.79 |
| Parasitemia | | 0.54 | 0.66 |
| G6PD (deficient vs. non) | 0.32 | 0.88 | 0.09 |
| Hemoglobin genotype | 0.50 | 0.71 | 0.63 |
| Hemoglobin count (g/dl) | | 0.80 | 0.08 |
| Geography (west vs. east) | 0.14 | 0.28 | 0.18 |
| Time of collection | 0.76 | 0.03* | 0.05* |

*significant at p < = 0.05.

doi:10.1371/journal.pntd.0004526.t001



Fig 5. Complexity of infection according to time of collection. The infections are separated in monoclonal and polyclonal infections (a) or according to the most likely number of clones (b). Numbers within the bars indicate sampling size for each category.

doi:10.1371/journal.pntd.0004526.g005

genotyping success between symptomatic and asymptomatic samples, we randomly subsampled the symptomatic genotypes (see <u>Materials and Methods</u> for details) and showed that symptomatic infections were more often polyclonal than asymptomatic infections (p = 0.04).

Genetic Diversity and Population Structure in Cambodian P. vivax

Population genetic analyses are complicated by the presence of multiple clones within an infection, which precludes reconstructing the genotype of each individual clone (i.e., the combination of alleles at different markers). We therefore limited our population genetic analyses to monoclonal (N = 25) and biclonal (N = 60) infections. Of these 85 parasites, 81 (95%) had a unique combination of alleles at the 100 initially targeted SNPs (i.e., a unique "barcode"), with a mean difference of 19.6 nucleotides between two infections. The infections caused by genetically undistinguishable parasites were always observed in the same location: two patients from Veurn Say were infected by parasites with the same genotypes at all 100 SNPs, as were a pair of patients from Rovieng and two pairs of patients from Pailin. These apparently similar infections at the same location could be caused by the same parasite infecting different individuals or by highly related parasites (e.g., recombinant clones). Overall, these findings indicated that most infections were caused by genetically very different parasites suggesting a high level of genetic diversity in the Cambodian *P. vivax* population.

We then assessed the population structure of Cambodian *P. vivax* using the entire DNA sequences of the 100 loci analyzed, treating them as multiallelic in the program STRUCTURE (using only genotypes at the 100 unlinked SNPs initially targeted yielded, qualitatively, identical results but with a lesser resolution, $R^2 = 0.83$, $p < 2.2 \times 10^{-16}$). To improve the power of our analyses, as well as facilitate the interpretation of the results, we included genotypes at the same loci from *P. vivax* parasites that have been sequenced for their entire genome [15,24–27].

STRUCTURE assignments showed clear population separation between Southeast Asian (Cambodian and Thai) and South American and African parasites ($p < 2.15 \times 10^{-6}$; Fig 6). Though Cambodian and Thai *P. vivax* showed ancestry from the same ancestral populations, they significantly differed in their coefficients of ancestry and appeared distinct (p = 0.0001; Fig 6). By contrast, we found no significant differences between parasites collected in western vs. eastern Cambodia (p = 0.09) or among sites (p = 0.09), suggesting that the Cambodian *P. vivax* population is little stratified geographically. Note that the parasite population also



Fig 6. Population stratification analyses showing assignment of the parasites into K = 2, 3, and 4populations based solely on their genotypes. Each vertical line displays the ancestry of one parasite (organized along x-axis) in each of the ancestral populations (represented by the different colors). Thin black lines separate districts in Cambodia. Individuals from Thailand and non-Southeast Asia are divided by thick vertical lines.

doi:10.1371/journal.pntd.0004526.g006

appeared to be genetically diverse in this analysis: the genomes did not fall into a single homogenous population, but rather were split into multiple populations (Fig 6). Additionally, we observed no association between the parasite genotypes and the age (p = 0.37), sex of the patient (p = 0.14), or with the sampling year (p = 0.79) (Table 1).

Discussion

A Quantitative and Customizable High-Throughput Genotyping Assay

Successful elimination of *P. vivax* will require monitoring how the parasite population responds to control measures to identify areas where elimination is less efficient due to logistical or biological reasons (e.g., high vector density or emergence of drug-resistant parasites). Several studies have used assessment of genetic diversity to measure *P. vivax* demographic

parameters (e.g., see [28–32]). However, due to the general paucity of genetic markers in *P. vivax*, most past studies have relied on a reduced number of microsatellite markers or SNPs, or on DNA sequences at highly variable surface protein antigens. Unfortunately, these markers only provide a partial perspective on the parasite population and could be biased by variations in mutation rates or the role of natural selection (e.g., positive selection on the locus investigated could, for example, be interpreted as evidence of population expansion). Here we described the development and implementation of a cost-efficient and high-throughput genotyping approach that is easily customizable and enables rigorous characterization of COI, genetic diversity and population stratification in the population of interest.

The present assay targets over 100 SNPs distributed across the vivax genome, which minimizes potential artifacts in demographic inferences caused by the effect of natural selection driving the evolution of some markers. In addition, genotyping by high-throughput sequencing provides two additional key advantages. First, since 100-300 bp surrounding each targeted SNP are sequenced, the assay can identify additional DNA polymorphisms upstream or downstream of the targeted SNP position (for a total of 274 SNPs in our study) and provide haplotype data (i.e., multi-SNPs genotypes) that are much more informative than single SNPs (e.g., see Fig 4). Second, since more than 100 sequences are generated for each SNP and each sample, the genotyping is quantitative and enables high sensitivity and rigorous identification of alleles even if they are carried by only a minority of the clones (a major difference compared to traditional genotyping approaches that have limited power to identify rare clones [33]). Note that, by simply changing the amplification primers used, the assay can be easily modified to target any locus of interest and could, for example, include the recently developed "P. vivax SNP barcode" [10] to enable direct comparison across studies. This flexibility is also important as it enables optimizing the experimental design to specific research questions. We chose here to analyze a large number of markers and stringent coverage cutoffs to rigorously investigate COI (but with a relatively high rate of missing data). Alternatively one could reduce the number of markers or the depth of coverage to obtain a more complete dataset (but with lower power to detect minor clones).

Complexity of Infection in Cambodian P. vivax

Overall, we observed a very high proportion of complex infections among Cambodian patients: 92% of all symptomatic infections showed clear evidence of having multiple clones. In 60 infections, we only detected a single variable position. This observation could possibly be explained by the presence of two closely related clones in the infection, similar to what has been described in *P. falciparum* [34,35]. However, most of the remaining 217 infections (72%) show numerous variable positions and likely represent instances of unrelated P. vivax clones simultaneous present in a given patient. This pervasive high COI contrasts with other studies that often show a large proportion of monoclonal infections (e.g., see [30-32]). While this discrepancy could be caused by lower parasite diversity or transmission rates compared to the situation in Cambodia, it is important to note that our estimates of COI are derived from sequencing numerous loci (instead of a few microsatellites or SNPs), which dramatically increases our power to differentiate clones and therefore to detect complex infections (Fig 4). In particular, our findings are consistent with previous studies that used next-generation sequencing approaches [15,36-38]. It is interesting to contrast these patterns of COI in P. vivax with studies of P. falciparum: for example, a recent study, including more than 200 Cambodian P. falciparum collected in the same areas as our P. vivax infections, showed that these samples were essentially clonal [39]. In this regard, our study confirms previous findings based on a small number of microsatellites that showed striking

differences in COI between sympatric *P. falciparum* and *P. vivax* infections collected in western Cambodia [40]. This difference is COI could reflect the biological differences between these species and, in particular, the role of the hypnozoites that might contribute to the elevated COI in *P. vivax* by releasing previously dormant parasites in the blood stream upon a new infection [41]. Alternatively, this difference could be caused by a recent decrease in *P. falciparum* diversity due to extreme drug pressure in this region (though both parasite species are present in the same areas and are often treated indiscriminately).

We made use of the large number of samples investigated (N = 489) to test the influence of several patient and demographic parameters on COI. None of the patient data (age, sex, hemoglobin level or genotype, G6PD deficiency or parasitemia) were significantly associated with COI. In particular, higher levels of parasites in the blood were not statistically associated with polyclonal infections. Interestingly, we did detect modest but significant differences in COI between symptomatic and asymptomatic infections, with slightly more clones present in symptomatic infections. This observation will need to be confirmed by future studies but is consistent with the results of evolutionary models [42,43]. One could speculate that in *P. vivax* infections, which typically have a low parasitemia and multiple clones per infection, a higher number of clones could complicate the task of the immune system by presenting a more diverse range of surface antigens and lead to a more severe infection. In P. falciparum, increased COI has been associated with both lower and higher risk of malaria symptoms [44– 47]. As more studies in both species become available, one will be able to rigorously address this interesting and clinically-important topic and test whether COI is associated with malaria symptoms and if the direction of this association is identical across *Plasmodium* species or in contrary, differs based on the specificities of each parasite (e.g., the differences in parasitemia and frequency of COI).

Some of the patient parameters investigated in our study had been previously reported to influence *P. vivax* infections [48–51]. For example, some hemoglobin variants [52,53] and G6PD genotypes [54,55] might confer some degree of protection against *P. vivax* infections. By contrast, our analyses showed that these parameters do not seem to influence the number of clones infecting a given patient. This apparent discrepancy between the influence of these parameters on *P. vivax* infection and on COI could indicate that the infection complexity does not arise from multiple independent infections but, instead, from either simultaneous infection by multiple clones (e.g., the mosquito might directly release multiple parasites in the blood) or by the relapse of dormant parasites from previous infections (see e.g., [56]).

Interestingly, the geographic origin of the patients did not correlate with COI despite the very different incidence of *P. vivax* in western and eastern provinces: the vast majority of reported *P. vivax* cases (80%) occurred in eastern Cambodia. By contrast, the year of collection was statistically associated with infection type: we observed that the proportion of monoclonal infections significantly increased in Cambodia from 2004 to 2013. This increase in monoclonal infection may reflect a general decline in the Cambodian *P. vivax* population, which may be indicative of the success of elimination programs [57] (e.g., see [58,59] for a similar pattern in *P. falciparum* in Senegal). Indeed, during this period, the number of malaria cases in Cambodia has been halved, dropping from 113,855 cases in 2004 down to 56,271 cases in 2014 (though a large component of this decrease can be attributed to a dramatic reduction in *P. falciparum* infections). This observation is encouraging for malaria control in Cambodia but analysis of additional samples collected more recently will be necessary to confirm this observation and to determine if this decrease is sustained over time.

Genetic Diversity and Population Stratification of P. vivax in Cambodia

We observed a high level of genetic diversity in Cambodian *P. vivax* as illustrated by the large numbers of nucleotide differences differentiating clones, with, for example, an average of 19.6 differences, out of the 100 targeted SNPs, between two clones (and up to 40 differences). This high diversity is also illustrated by the results of our population structure analyses: most parasite genomes do not cluster homogeneously into one population, but instead display complex patterns of ancestry from multiple populations (Fig 6).

Interestingly, this extensive genetic diversity seemed to be distributed throughout Cambodia without clear demarcations: structure analysis failed to reveal any evidence of population stratification within Cambodia (Fig 6) despite the existence of a low transmission zone separating our western and eastern sampling locations (Fig 1). Similar results have been reported for P. falciparum populations [60]. While the 100 loci sequenced provide sufficient information to rigorously differentiate parasites from Southeast Asia and the rest of the world, and to detect significant differences between Cambodian P. vivax and (western) Thai P. vivax, it is possible that more genetic markers (or whole genome sequence data) would be necessary to identify subtle population differences within Cambodia. Regardless, two possible demographic scenarios could explain the lack of strong population stratification. First, it is possible that the current Cambodian P. vivax population derive from a single diverse ancestral population that was only recently split into multiple separated populations, and that these have yet to differentiate significantly. Alternatively, our observation could be consistent with a high level of gene flow between different Cambodian P. vivax populations, resulting in high level of admixture (which could explain the multiple ancestry patterns shown on Fig 6). However, differentiating these hypotheses would require extensive whole genome sequencing data that would enable rigorously quantifying gene flow, population stratification and the demographic histories of the parasite populations.

Conclusion

Despite progress in malaria control in general, vivax malaria remains an important public health concern in many endemic countries. The lack of tools to assess the efficacy of ongoing control efforts hampers the rapid implementation of alternative or complementary approaches in the field. We described here a novel genotyping assay that can interrogate over 100 SNPs across the *P. vivax* genome, in a rapid, high-throughput and cost-efficient manner and can be tailored to specific polymorphisms or loci of interest. Using this approach, we were able to characterize geographical and temporal variations in *P. vivax* genetic diversity in Cambodia. Our analyses revealed that the proportion of monoclonal infections is increasing in Cambodia, which may be an indication of the success of the on-going control measures. However, our analyses also suggest that gene flow among Cambodian *P. vivax* is important, which could be problematic if antimalarial drug resistant alleles arise in one site as they could then inevitably spread throughout the region. Overall, this high-throughput genotyping technique adds new options to our existing toolkit for malaria control by providing a rigorous framework to assess how parasite populations respond to disease control strategies.

Supporting Information

S1 Table. Number of (*a*) symptomatic and (*b*) asymptomatic samples included in this study according to their geographical origin. (PDF)

S2 Table. Primer information for multiplexing assay. Bold denote SNP positions used for our final analyses.

(PDF)

S1 Fig. Comparison of the allele frequencies estimated in mono- and "biclonal" infections and polyclonal infections. Each point represents one of the initially targeted SNP and is displayed based on its average reference allele frequency (RAF) in monoclonal and biclonal infections (x-axis) and in polyclonal infections (y-axis). (TIF)

S2 Fig. Comparison of the genotyping results between duplicated samples. Each plot corresponds to a different infection, with monoclonal samples being shown in (*a*) and (*b*), and polyclonal in (*c*) and (*d*). Each diamond represents one SNP and is displayed according to its allele frequency in the duplicated samples. (TIF)

S3 Fig. Complexity of infection in our symptomatic patient samples. The infections are separated in monoclonal and polyclonal infections (*a*) and according to the most likely number of clones (*b*). Numbers indicate sampling size for each category. (TIF)

S4 Fig. Complexity of infection across years in *(a)* Pailin and *(b)* Veurn Say. The proportion of monoclonal infections is shown in black and the polyclonal infections in grey. The figures indicate the actual number of samples in each category. (TIF)

Acknowledgments

We thank all patients and healthcare workers involved in this study and the staff of the Malaria Molecular Epidemiology Unit at the Institut Pasteur and of the National Center for Parasitology, Entomology and Malaria Control in Cambodia for their collaboration and sample collection.

Author Contributions

Conceived and designed the experiments: DS DM. Performed the experiments: LRF. Analyzed the data: LRF DS. Contributed reagents/materials/analysis tools: JP SK LD PAZ DM DS. Wrote the paper: LRF JP DM DS.

References

- Battle KE, Gething PW, Elyazar IRF, Moyes CL, Sinka ME, Howes RE, et al. (2012) The global public health significance of *Plasmodium vivax*. Adv Parasit 80. Available: <u>http://dx.doi.org/10.1016/B978-0-12-397900-1.00001-3</u>
- Price RN, Tjitra E, Guerra CA, Yeung S, White NJ, Anstey NM (2007) Vivax malaria: neglected and not benign. Am. J Trop Med Hyg 77: 79–87. PMID: <u>18165478</u>
- Noulin F, Borlon C, Van Den Abbeele J, D'Alessandro U, Erhart A (2013) 1912–2012: a century of research on *Plasmodium vivax in vitro* culture. Trends Parasitol 29: 286–294. Available: doi: <u>http://dx. doi.org/10.1016/j.pt.2013.03.012</u> PMID: 23623759
- 4. Basco LK, Le Bras J (1994) Short-term *in vitro* culture of *Plasmodium vivax* and *P. ovale* for drug-susceptibility testing. Parasitol Res 80: 262–264. PMID: <u>8036243</u>
- Russell BM, Udomsangpetch R, Rieckmann KH, Kotecka BA, Coleman RE, Sattabongkot J (2003) Simple in vitro assay for determining the sensitivity of *Plasmodium vivax* isolates from fresh human blood to antimalarials in areas where *P. vivax* is endemic. Antimicrob Agents Chemother 47: 170–173. PMID: <u>12499187</u>

- Russell B, Malleret B, Suwanarusk R, Anthony C, Kanlaya S, Lau YL, et al. (2013) Antimicrob Agents Chemother. 2013 Oct; 57(10): 5170–5174. Field-based flow cytometry for *Ex Vivo* characterization of *Plasmodium vivax* and *P. falciparum* antimalarial sensitivity. Antimicrob Agents Chemother 57: 5170– 5174. doi: 10.1128/AAC.00682-13 PMID: 23877705
- Imwong M, Nair S, Pukrittayakamee S, Sudimack D, Williams JT, Mayxay M, et al. (2007) Contrasting genetic structure in *Plasmodium vivax* populations from Asia and South America. Int J Parasitol 37: 1013–1022. PMID: <u>17442318</u>
- Gunawardena S, Karunaweera ND, Ferreira MU, Phone-Kyaw M, Pollack RJ, Alifrangis M, et al. (2012) Geographic structure of *Plasmodium vivax*: microsatellite analysis of parasite populations from Sri Lanka, Myanmar, and Ethiopia. Am J Trop Med Hyg 82: 235–242.
- Koepfli C, Rodrigues PT, Antao T, Orjuela-Sánchez P, Van den Eede P, Gamboa D, et al. (2015) *Plasmodium vivax* diversity and population structure across four continents. PLoS Negl Trop Dis 9: e0003872. doi: 10.1371/journal.pntd.0003872 PMID: 26125189
- Baniecki ML, Faust AL, Schaffner SF, Park DJ, Galinsky K, Daniels RF, et al. (2015) Development of a single nucleotide polymorphism barcode to genotype *Plasmodium vivax* infections. PLoS Negl Trop Dis 9: e0003539. doi: 10.1371/journal.pntd.0003539 PMID: 25781890
- Orjuela-Sánchez P, Karunaweera ND, da Silva-Nunes M, da Silva NS, Scopel KKG, Gonçalves RM, et al. (2010) Single-nucleotide polymorphism, linkage disequilibrium and geographic structure in the malaria parasite *Plasmodium vivax*: prospects for genome-wide association studies. BMC Genetics 11: 65. doi: 10.1186/1471-2156-11-65 PMID: 20626846
- Cui L, Escalante AA, Imwong M, Snounou G (2003) The genetic diversity of *Plasmodium vivax* populations. Trends Parasitol 19: 220–226. PMID: <u>12763428</u>
- Venkatesan M, Amaratunga C, Campino S, Auburn S, Koch O, Lim P, et al. (2012) Using CF11 cellulose columns to inexpensively and effectively remove human DNA from *Plasmodium falciparum*infected whole blood samples. Malaria J 11: 41.
- Melnikov A, Galinsky K, Rogov P, Fennell T, Tyne DV, Russ C, et al. (2011) Hybrid selection for sequencing pathogen genomes from clinical samples. Genome Biol 12: R73. doi: <u>10.1186/gb-2011-</u> <u>12-8-r73</u> PMID: <u>21835008</u>
- Chan ER, Menard D, David PH, Ratsimbasoa A, Kim S, Chim P, et al. (2012) Whole genome sequencing of field isolates provides robust characterization of genetic diversity in *Plasmodium vivax*. PLoS Negl Trop Dis 6: e1811. doi: <u>10.1371/journal.pntd.0001811</u> PMID: <u>22970335</u>
- Untergrasser A, Cutcutache I, Koressaar T, Ye J, Faircloth BC, Remm M, et al. (2012) Primer3—new capabilities and interfaces. Nucleic Acids Res 40: e115. PMID: <u>22730293</u>
- Masella AP, Bartram AK, Truszkowski JM, Brown DG, Neufeld JD (2012) PANDAseq: paired-end assembler for Illumina sequences. BMC Bioinformatics 13: 31. doi: <u>10.1186/1471-2105-13-31</u> PMID: <u>22333067</u>
- Carlton JM, Adams JH, Silva JC, Bidwell SL, Lorenzi H, Caler E, et al. (2008) Comparative genomics of the neglected human malaria parasite *Plasmodium vivax*. Nature 455: 757–763. doi: <u>10.1038/</u> nature07327 PMID: 18843361
- Langmead B, Salzberg SL (2012) Fast gapped-read alignment with Bowtie 2. Nat Methods 9: 357– 359. doi: 10.1038/nmeth.1923 PMID: 22388286
- Tamura K, Stecher G, Peterson D, Filipski A, Kumar S (2013) MEGA6: molecular evolutionary genetics analysis version 6.0. Mol Bio Evol 30: 2725–2729.
- Galinsky K, Valim C, Salmier A, de Thoisy B, Musset L, Legrand E, et al. (2015) COIL: a methodology for evaluating malarial complexity of infection using likelihood from single nucleotide polymorphism data. Malar J 14: 4. doi: 10.1186/1475-2875-14-4 PMID: 25599890
- 22. Khim N, Benedet C, Kim S, Kheng S, Siv S, Leang R, et al. (2013) G6PD deficiency in *Plasmodium falciparum* and *Plasmodium vivax* malaria-infected Cambodian patients. Malaria J 12: 171.
- Pritchard J K, Wen W (2004) Documentation for STRUCTURE software: version 2.2. University of Chicago, Chicago.
- Auburn S, Marfurt J, Maslen G, Campino S, Rubio VR, Manske M, et al. (2013) Effective preparation of *Plasmodium vivax* field isolates for high-throughput whole genome sequencing. PLoS ONE 8: e53160. doi: 10.1371/journal.pone.0053160 PMID: 23308154
- Neafsey DE, Galinsky K, Jiang RHY, Young L, Sykes SM, Saif S, et al. (2012) The malaria parasite *Plasmodium vivax* exhibits greater genetic diversity than *Plasmodium falciparum*. Nat Genet 44: 1046– 1050. doi: <u>10.1038/ng.2373</u> PMID: <u>22863733</u>
- Hester J, Chan ER, Menard D, Mercereau-Puijalon O, Barnwell J, Zimmerman PA, et al. (2013) *De novo* assembly of a field isolate genome reveals novel *Plasmodium vivax* erythrocyte invasion genes. PLoS Negl Trop Dis 7: e2569. doi: 10.1371/journal.pntd.0002569 PMID: 24340114

- Menard D, Chan ER, Benedet C, Ratsimbasoa A, Kim S, Chim P, et al. (2013) Whole genome sequencing of field isolates reveals a common duplication of the Duffy Binding Protein gene in Malagasy *Plasmodium vivax* strains. PLoS Negl Trop Dis 7: e2489. doi: <u>10.1371/journal.pntd.0002489</u> PMID: <u>24278487</u>
- Zakeri S, Abouie-Mehrizi A, Djadid ND, Snounou G (2006) Circumsporozoite protein gene diversity among temperate and tropical *Plasmodium vivax* isolates from Iran. Trop Med Int Health 11: 729–737. PMID: <u>16640626</u>
- Moon S, Lee H, Kim J, Na B, Cho S, Lin K, et al. (2009) High frequency of genetic diversity of *Plasmo-dium vivax* field isolates in Myanmar. Acta Tropica 109: 30–36. doi: <u>10.1016/j.actatropica.2008.09.006</u> PMID: <u>18851938</u>
- **30.** Van den Eede P, Van der Auwera G, Delgado C, Huyse T, Soto-Calle VE, Gamboa D, et al. (2010) Multilocus genotyping reveals high heterogeneity and strong local population structure of the *Plasmodium vivax* population in the Peruvian Amazon. Malaria J 9: 151.
- Chenet SM, Schneider KA, Villegas L, Escalante AA (2012) Local population structure of *Plasmodium*: impact on malaria control and elimination. Malaria J 11: 412.
- Abdulla NR, Barber BE, William T, Norahmad NA, Satsu UR, Muniandy PK, et al. (2013) *Plasmodium vivax* population structure and transmission dynamics in Sabah Malaysia. PLoS ONE 8: e82553. doi: 10.1371/journal.pone.0082553 PMID: 24358203
- Su X, Jiang H, Yi M, Mu J, Stephens RM (2009) Large-scale genotyping and genetic mapping in *Plasmodium* parasites. Korean J Parasitol 47 83–91. doi: <u>10.3347/kjp.2009.47.2.83</u> PMID: <u>19488413</u>
- Nair S, Nkhoma SC, Serre D, Zimmerman PA, Gorena K, Daniel BJ, et al. (2014) Single-cell genomics for dissection of complex malaria infections. Genome Res 24: 1028–1038. doi: <u>10.1101/gr.168286.113</u> PMID: <u>24812326</u>
- Nkhoma SC, Nair S, Cheeseman IH, Rohr-Allegrini C, Singlam S, Nosten F, et al. (2012) Close kinship within multiple-genotype malaria parasite infections. Proc R Soc B 279: 2589–2598. doi: <u>10.1098/rspb.</u> 2012.0113 PMID: 22398165
- Juliano JJ, Porter K, Mwapasa V, Sem R, Rogers WO, Ariey F, et al. (2010) Exposing malaria in-host diversity and estimating population diversity by capture-recapture using massively parallel pyrosequencing. PNAS 107: 20138–20143. doi: <u>www.pnas.org/cgi/doi/10.1073/pnas.1007068107</u> PMID: 21041629
- 37. Bailey JA, Mvalo T, Aragam N, Weiser M, Congdon S, Kamwendo D, et al. (2012) Use of massively parallel pyrosequencing to evaluate the diversity of and selection on *Plasmodium falciparum csp* T-cell epitopes in Lilongwe, Malawi. J Infect Dis 206: 580–587. doi: 10.1093/infdis/jis329 PMID: 22551816
- Taylor SM, Parobek CM, Aragam N, Ngasala BE, Mårtensson A, Meshnick SR, et al. (2013) Pooled deep sequencing of *Plasmodium falciparum* isolates: an efficient and scalable tool to quantify prevailing malaria drug resistance genotypes. J Infect Dis 208: 1998–2006. doi: <u>10.1093/infdis/jit392</u> PMID: 23908494
- Miotto O, Almagro-Garcia J, Manske M, Macinnis B, Campino S, Rockett KA, et al. (2013) Multiple populations of artemisinin-resistant *Plasmodium falciparum* in Cambodia. Nat Genet 45(6): 648–655. doi: 10.1038/ng.2624 PMID: 23624527
- 40. Orjuela-Sánchez P, Sá JM, Brandi MC, Rodrigues PT, Bastos MS, Amaratunga C, et al. (2013) Higher microsatellite diversity in *Plasmodium vivax* than in sympatric *Plasmodium falciparum* populations in Pursat, Western Cambodia. Exp Parasitol 134: 318–326. doi: <u>10.1016/j.exppara.2013.03.029</u> PMID: <u>23562882</u>
- 41. Shanks GD, White NJ (2013) The activation of vivax malaria hypnozoites by infectious diseases. Lancet Infect Dis 13(10): 900–906. doi: 10.1016/S1473-3099(13)70095-1 PMID: 23809889
- de Roode JC, Pansini R, Cheesman SJ, Helinski MEH, Huijben S, Wargo AR, et al. (2005) Virulence and competitive ability in genetically diverse malaria infections. Proc Natl Acad Sci USA 102: 7624– 7628. PMID: <u>15894623</u>
- Bell AS, De Roode JC, Sim D, Read AF (2006) Within-host competition in genetically diverse malaria infections: parasite virulence and competitive success. Evolution 60: 1358–1371. PMID: <u>16929653</u>
- 44. al-Yaman F, Genton B, Reeder JC, Anders RF, Smith T, Alpers MP (1997) Reduced risk of clinical malaria in children infected with multiple clones of *Plasmodium falciparum* in a highly endemic area: a prospective community study. Trans R Soc Trop Med Hyg 91: 602–605. PMID: 9463681
- 45. Beck HP, Felger I, Huber W, Steiger S, Smith T, Weiss N, et al. (1997) Analysis of multiple *Plasmodium falciparum* infections in Tanzanian children during the phase III trial of the malaria vaccine SPf66. J Infect Dis 175: 921–926. PMID: 9086150

- 46. Owusu-Agyei S, Smith T, Beck HP, Amenga-Etego L, Felger I (2002) Molecular epidemiology of *Plasmodium falciparum* infections among asymptomatic inhabitants of a holoendemic malarious area in northern Ghana. Trop Med Int Health 7: 421–428. PMID: <u>12000651</u>
- 47. Henning L, Schellenberg D, Smith T, Henning D, Alonso P, Tanner M, et al. (2004) A prospective study of *Plasmodium falciparum* multiplicity of infection and morbidity in Tanzanian children. Trans R Soc Trop Med Hyg 98: 687–694. PMID: 15485698
- Ntoumi F, Contamin H, Rogier C, Bonnefoy S, Trape JF, Mercereau-Puijalon O (1995) Age-dependent carriage of multiple *Plasmodium falciparum* merozoite surface antigen-2 alleles in asymptomatic malaria infections. Am J Trop Med Hyg 52: 81–88. PMID: 7856831
- 49. Smith T, Beck HP, Kitua A, Mwankusye S, Felger I, Fraser-Hurt N, et al. (1999) Age dependence of the multiplicity of *Plasmodium falciparum* infections and of other malariological indices in an area of high endemicity. Trans R Soc Trop Med Hyg 93: 15–20.
- Felger I, Smith T, Edoh D, Kitua A, Alonso P, Tanner M, et al. (1999) Multiple Plasmodium falciparum infections in Tanzanian infants. Trans R Soc Trop Med Hyg 93: 29–34. PMID: <u>10450423</u>
- Guerra-Neira A, Rubio JM, Royo JR, Ortega JC, Auñón AS, Diaz PB, et al. (2006) Plasmodium diversity in non-malaria individuals from the Bioko Island in Equatorial Guinea (West Central-Africa). Int J Health Geogr 5: 27. PMID: <u>16784527</u>
- 52. Kar S, Seth S, Seth PK (1992) Prevalence of malaria in Ao Nagas and its association with G6PD and HbE. Hum Biol 64: 187–197. PMID: <u>1559689</u>
- Taylor SM, Parobek CM, Fairhurst RM (2012) Haemoglobinopathies and the clinic epidemiology of malaria: a systematic review and meta-analysis. Lancet Infect Dis 12: 457–68. doi: <u>10.1016/S1473-3099(12)70055-5</u> PMID: <u>22445352</u>
- Louicharoen C, Patin E, Paul R, Nuchprayoon I, Witoonpanich B, Peerapittayamongkol C, et al. (2009) Positively selected G6PD-Mahidol mutation reduces *Plasmodium vivax* density in Southeast Asians. Science 326: 1546–1548. doi: 10.1126/science.1178849 PMID: 20007901
- Leslie T, Briceño M, Mayan I, Mohammed N, Klinkenberg E, Sibley CH, et al. (2010) The impact of phenotypic and genotypic G6PD deficiency on risk of *Plasmodium vivax* infection: a case-control study amongst Afghan refugees in Pakistan. PLoS Med. 7: e1000283. doi: <u>10.1371/journal.pmed.1000283</u> PMID: <u>20520804</u>
- Imwong M, Snounou G, Pukrittayakamee S, Tanomsing N, Kim JR, Nandy A, et al. (2007) Relapses of *Plasmodium vivax* infection usually result from activation of heterologous hypnozoites. JID 195: 927– 933. PMID: <u>17330781</u>
- Daniels R, Rice BL, Daniels NM, Volkman SK, Hartl DL (2015) The utility of genomic data for *Plasmo-dium vivax* population surveillance. Pathog Glob Health 109: 153–161. Available: http://dx.doi.org/10.1179/2047773215Y.00000001 doi: http://dx.doi.org/10.1179/2047773215Y.000000014 doi: http://dx.doi.org/10.1179/2047773215Y.000000014 doi: http://dx.doi.org/10.1179/2047773215Y.000000014 doi: http://dx.doi.org/10.1179/2047773215Y.000000014 doi: http://dx.doi.org/10.1179/2047773215Y.0000000014 DMID: http://dx.doi.org/10.1179/2047773215Y.0000000014 DMID: http://dx.doi.org/10.1179/2047773215Y.0000000014 DMID: http://dx.doi.018 DMID: http://dx.doi.018 DMID: http://dx.doi.org/10.1179/2047773215Y.000000014 DMID: http://dx.doi.018 DMID: http://dx.doi.org/10.11
- Nkhoma SC, Nair S, Al-Saai S, Ashley E, McGready R, Phyo AP, et al. (2013) Population genetic correlates of declining transmission in a human pathogen. Mol Ecol 22: 273–285. doi: <u>10.1111/mec.12099</u> PMID: 23121253
- 59. Daniels R, Chang H-H, Se ne PD, Park DC, Neafsey DE, Schaffner SF, et al. (2013) Genetic surveillance detects both clonal and epidemic transmission of malaria following enhanced intervention in Senegal. PLoS ONE 8: e60780. doi: 10.1371/journal.pone.0060780 PMID: 23593309
- Miotto O, Amato R, Ashley EA, MacInnis B, Almagro-Garcia J, Amaratunga C, et al. (2015) Genetic architecture of artemisinin-resistant *Plasmodium falciparum*. Nat Genet 47: 226–234. doi: <u>10.1038/ng.</u> <u>3189</u> PMID: <u>25599401</u>